\begin{thebibliography}{10}

\bibitem{n3logic}
{\sc Berners-Lee, T., Connolly, D., Kagal, L., Scharf, Y., and Hendler, J.}
\newblock N3logic: A logical framework for the world wide web.
\newblock {\em Journal of Theory and Practice of Logic Programming\/} (2007).

\bibitem{acl-oracle}
{\sc Byun, J.-W.}
\newblock Access control features in oracle.
\newblock
  \url{http://www.cs.purdue.edu/homes/ninghui/courses/Spring05/lectures/Oracle%
.pdf}, April 2005.

\bibitem{Cathey03misusedetection}
{\sc Cathey, R., Ma, L., Goharian, N., and Grossman, D.}
\newblock Misuse detection for information retrieval systems.
\newblock In {\em ACM 12th Conference on Information and Knowledge
  Management\/} (2003).

\bibitem{demids}
{\sc Chung, C.~Y., Gertz, M., and Levitt, K.}
\newblock Demids: A misuse detection system for database systems.
\newblock In {\em In Third International IFIP TC-11 WG11.5 Working Conference
  on Integrity and Internal Control in Information Systems\/} (1999), Kluwer
  Academic Publishers, pp.~159--178.

\bibitem{rbac}
{\sc Ferraiolo, D.~F., and Kuhn, D.~R.}
\newblock Role-based access controls.
\newblock In {\em 15th National Computer Security Conference\/} (October 1992),
  pp.~554--563.

\bibitem{oracle}
{\sc Hale, L.~P., and Levinger, J.}
\newblock Oracle database security guide, 10g release 1.
\newblock
  \url{http://www.stanford.edu/dept/itss/docs/oracle/10g/network.101/b10773/ti%
tle.htm}, December 2003.

\bibitem{rein}
{\sc Kagal, L., Berners-Lee, T., Connolly, D., and Weitzner, D.}
\newblock Using semantinc web technologies for policy management on the web.
\newblock In {\em 21st National Conference on Artificial Intelligence (AAAI)\/}
  (July 2006).

\bibitem{air}
{\sc Kagal, L., Hanson, C., and Weitzner, D.}
\newblock Using dependency tracking to provide explanations for policy
  management.
\newblock {\em IEEE Policy 2008\/} (2008).

\bibitem{Kamra}
{\sc Kamra, A., Terzi, E., and Bertino, E.}
\newblock Detecting anomalous access patterns in relational databases.
\newblock {\em VLDB Journal: The International Journal on Very Large Data
  Bases\/} (2007).

\bibitem{rulebased}
{\sc Li, H., Zhang, X., Wu, H., and Qu, Y.}
\newblock Design and application of rule based access control policies.
\newblock In {\em ISWC Workshop on Semantic Web and Policy\/} (2005),
  pp.~34--41.

\bibitem{mysql}
{\sc {MySQL A.B.}, and {Sun Microsystems}}.
\newblock Mysql 6.0 reference manual.
\newblock \url{http://dev.mysql.com/doc/refman/6.0/en/index.html}, 2009.

\bibitem{spasql}
{\sc Prud'hommeaux, E.}
\newblock Spasql: Sparql support in mysql.
\newblock \url{http://www.w3.org/2005/05/22-SPARQL-MySQL/XTech}, June 2007.

\bibitem{Qiu85trustedcomputer}
{\sc Qiu, L., Zhang, Y., Wang, F., Kyung, M., and Mahajan, H.~R.}
\newblock Trusted computer system evaluation criteria.
\newblock In {\em National Computer Security Center\/} (1985).

\bibitem{rbac-commercial}
{\sc Ramaswamy, R., and S, R.}
\newblock Role-based access control features in commercial database management
  systems.
\newblock In {\em In Proceedings of 21st NIST-NCSC National Information Systems
  Security Conference\/} (1998), pp.~503--511.

\bibitem{creative-commons}
{\sc Seneviratne, O., Kagal, L., Weitzner, D., Abelson, H., Berners-Lee, T.,
  and Shadbolt, N.}
\newblock Detecting creative commons license violations on images on the world
  wide web.
\newblock In {\em WWW2009\/} (April 2009).

\bibitem{datafly}
{\sc Sweeney, L.}
\newblock Guaranteeing anonymity when sharing medical data: the datafly system.
\newblock In {\em AMIA Annual Fall Symposium\/} (1997).

\bibitem{kanon2}
{\sc Sweeney, L.}
\newblock Achieving k-anonymity privacy protection using generalization and
  suppression.
\newblock {\em International Journal on Uncertainty, Fuzziness, and
  Knowledge-Based Systems 10 (5)\/} (2002), 571--588.

\bibitem{kanon}
{\sc Sweeney, L.}
\newblock k-anonymity: A model for protecting privacy.
\newblock {\em International Journal on Uncertainty, Fuzziness, and
  Knowledge-Based Systems 10 (5)\/} (2002), 555--570.

\bibitem{dac-db}
{\sc Thomas, R.~K., and Sandhu, R.~S.}
\newblock Discretionary access control in object-oriented databases: Issues and
  research directions.
\newblock In {\em National Computer Security Conference\/} (September 1993),
  pp.~63--74.

\bibitem{sparql}
Sparql query language for rdf.
\newblock \url{http://www.w3.org/TR/rdf-sparql-query/}, January 2008.

\bibitem{info-account}
{\sc Weitzner, D.~J., Abelson, H., Berners-Lee, T., Feigenbaum, J., Hendler,
  J., and Sussman, G.~J.}
\newblock Information accountability.
\newblock {\em Communications of the ACM\/} (June 2008).

\end{thebibliography}

\chapter{Introduction}

Databases are a common way of storing data, and SQL is arguably the most popular variety of database in production as of this writing. The most popular SQL databases implementations, Oracle and MySQL, offer a rudimentary set of security options. SQL has a table-centric view of data, and a UNIX-like security model. \cite{mysql} A database super-user can create individual tables, create user accounts, and grant or deny access to certain tables to certain users.

The existing state of the art in database security is administrator-centric and based on security models that are decades old. While adequate for many applications, these systems have limitations. They provide an ``all or nothing'' model of security, with no ability to show which rules provided an output or a specialized, data centric policy system. More information about the conclusions that a policy system reached is helpful. It promotes transparency, helps find errors, helps keep users honest, and helps to promote more fine-grained access policies. I expand on these systems with the use of Semantic Web technologies. In particular, I offer policy assurance, which provides a guarantee that a database query was or was not in compliance with a policy. Policy assurance provides an exact description of why a query is or is not compliant. 

To securely grant access to a piece of information, data security engineers typically ask the following questions, or make assumptions as necessary:
\begin{itemize}
    \item Who are you?
    \item What information are you trying to see?
    \item Why do you want to see this information?
\end{itemize}

Electronic systems traditionally utilize a few common design patterns in their implementations of security primitives. Discretionary access control \cite{dac-db} is the most familiar approach, and the basis of the UNIX security model. A user or super-user can expressly allow or deny other users certain permissions to certain pieces of data. Mandatory access control \cite{Qiu85trustedcomputer} requires explicit permission for any action, granted by an outside body; for a time, this was a popular design pattern for government systems containing classified data. Role-based access control \cite{rbac} groups users into specific roles, which themselves have specific permissions. Rule based access control \cite{rulebased} allows an administrator to create more specific policies for governing access.

This thesis documents a database security system built using Semantic Web technologies and design principles. I argue that a query to a database contains a substantial amount of information about what a user is trying to access, and why they wish to do so. I design and implement a system that implements rudimentary database security, and provides explanations for the policies it enforces. Our model user is ``honest but curious'': a trusted, well-intentioned employee of a government agency who is using a database, provided under numerous policies and agreements, to the user's agency.

The thesis introduces some key contributions to the field. The SPARQL to N3 conversion tool and related ontology are novel. The definitions of template policies, and tools for creating them, reduce policy based security to a few primitives. Policy assurance provides more transparency to users and administrators alike.

I approached this project in two phases. In phase one, I used SWObjects to convert SPARQL queries to an RDF serialization. I wrote low level AIR policies to check the compliance of these SPARQL queries, and demonstrated some test cases. In phase two, I created an abstraction for these SPARQL queries, and provided a meta level that removed dependence on the data structure.

\section{Motivating Example}

\begin{quote}
\emph{``Policy Assurance'' is a set of technical mechanisms that enable effective and accountable information sharing and usage.}\cite{info-account}
\end{quote}

``Policy assurance'' is a process by which we can be anywhere from reasonably to absolutely certain that actions in a system are in compliance with an existing policy. To validate the claim of compliance (or noncompliance), there is some data set which validates the claim.

As a sample scenario, we consider a database that contains highly sensitive data and, as a result, is only accessible to a limited number of people. For this possible database, we assume that the database administrator is not the database owner, and thus, cannot see either the contents of the database or the queries that external parties make to the database. The database owner may only see the information in the database, but not the queries made to it. Policy assurance allows the database owner and database administrator to verify that every query made to the database was in compliance with the given policies, without exposing any sensitive information.

The key point of policy assurance is that, if an external party knows the policies and has access to enough information about each policy check, the external party can verify the compliance of the policies. If there is a guarantee that every single access is logged, then an external party can validate a claim of complete compliance. This promotes transparency, by allowing verification, while providing security, which we might define as ``minimize the information required to permit verification.''

\subsection{Sample Usage Scenario}

In this thesis, we will consider a simple usage scenario of an administrator who wishes to guard data in an RDF data set which users access using SPARQL queries. In this scenario, users may not access location information about entities in the datbase; specifically, they may not retrieve or use address information in any form. The administrator may not see the queries that users are making, for security reasons, but must be certain that \emph{every single query} made to the data set is compliant with the policy about addresses.

The system described in this thesis is well suited to this task. In later sections, we will describe how the administrator would create a policy to enforce this, and how the administrator would check some sample queries manually. We will then describe how this system could be configured to perform this task automatically, providing policy assurance for the administrator and security for the users.

\section{System Components}

In this thesis, I present a system that provides policy assurance for SPARQL databases by checking whether incoming queries conform to authorization policies written in AIR. The system uses query based security to draw its conclusions. This system has several components:

\begin{itemize}
	\item A translator, which generates an N3 output from a SPARQL query. This output serves as an input to a reasoner.
	\item An ontology which serves to define the N3 output of the translator.
	\item A number of pre-defined policy templates in AIR, which capture the behavior of some of the most common policy design patterns. Some policies support reasoning over a log of previous queries.
	\item An algorithm for automatically generating AIR policies by combining user inputs with policy templates.
	\item A Web interface for performing SPARQL to N3 translation.
	\item A Web interface for generating policies.
	\item A Web interface for checking queries against policies.
\end{itemize}

All of the Web interfaces in this thesis work with Tabulator's Justification UI, for user-friendly viewing of queries and policies.


\begin{figure}
\centering
\includegraphics{system-drawing-1}
\caption{Architecture of our policy assurance reasoner, demonstrating separation from the RDBMS.}
\label{system-drawing}
\end{figure}

\section{Outline}

The remainder of this thesis is structured as follows. The following chapter provides a demonstration of the system, elaborates on the topic of policy assurance, and provides an overview of how this system is useful in implementing policy assurance. Following the overview, I discuss the design and implementation of each component of the system in detail, with a discussion of the design assumptions and trade-offs necessary for implementation. Then, I explore the performance of the reasoner. I then look at future directions for this work, and draw comparisons with other work in the field, before coming to a conclusion.

The appendix chapters provide background information. I discuss the ongoing Semantic Web initiative and the technologies which provide the infrastructure for this work. I offer code samples and useful information about the project code.